PSYCHOLOGICAL FACTORS AFTER STROKE: ARE THEY STABLE OVER TIME?

Melloney L. M. WIJENBERG, $MSc^{1,2}$, Caroline M. VAN HEUGTEN, PhD^{1-3} , Maria L. VAN MIERLO, $PhD^{4,5}$, Johanna M. A. VISSER-MEILY, PhD^4 and Marcel W. M. POST, $PhD^{4,6}$

From the ¹Department of Neuropsychology and Psychopharmacology, Faculty of Psychology and Neuroscience, Maastricht University, Maastricht, ²Limburg Brain Injury Center, ³Department of Psychiatry and Neuropsychology, School for Mental Health and Neuroscience, Maastricht University, Maastricht, ⁴Center of Excellence for Rehabilitation Medicine, Brain Center Rudolf Magnus, University Medical Center Utrecht and De Hoogstraat Rehabilitation, Utrecht, ⁵Clinical Research Department, Rijnstate Hospital, Arnhem and ⁶Department of Rehabilitation Medicine, University of Groningen, University Medical Center Groningen, Groningen, The Netherlands

Objective: Psychological factors influence stroke outcomes, such as participation and quality of life. Although important for clinical practice, not much is known about the temporal stability of these factors. This study explored whether psychological factors are stable post-stroke.

Methods: Prospective longitudinal cohort study. The following psychological factors were assessed using self-report questionnaires at 2 months and at 2 years post-stroke: proactive coping, self-efficacy, extraversion, optimism, passive coping, neuroticism and pessimism. Changes over time, associations and dimensions among psychological factors were considered.

Results: Data for 324 participants were available. Only passive coping scores showed no change between 2 months and 2 years post-stroke. Participants showed less proactive coping, lower self-efficacy, less extraversion, less optimism, more neuroticism and more pessimism over time. All but one inter-correlation of psychological factors, r = [-0.14; 0.71], and all correlations over time, r = [0.42-0.64], were significant. At both time-points, the psychological factors clustered into an "adaptive psychological factor" (proactive coping, self-efficacy, extraversion) and a "maladaptive psychological factor" (passive coping, neuroticism).

Conclusion: Across all psychological factors, changes toward less favourable scores were found. Clinicians should pay attention to adaptive and maladaptive psychological factors among stroke patients during long-term care.

Key words: stroke; psychological factors; rehabilitation; longitudinal studies.

Accepted Aug 7, 2018; Epub ahead of print xx, 2018

J Rehabil Med 2018; 50: 00-00

Correspondence address: Caroline M. van Heugten, NP&PP, FPN, Maastricht University, PO Box 616, 6200 MD Maastricht, The Netherlands. E-mail: caroline.vanheugten@maastrichtuniversity.nl

Stroke is a major health problem due to its high prevalence, disabling consequences and risk of recurrence (1). According to the International Classification of Disability, Functioning and Health (ICF), functioning and health are influenced by personal

LAY ABSTRACT

Every person has a distinct personality. Personality traits may influence how people recover from stroke. This study followed 324 patients in the first 2 years after stroke and examined whether these personality traits are stable over time. We found changes to less favourable outcomes, such as less extraversion, less optimism and more neuroticism. Clinicians should be aware of these changes and stimulate the use of more adaptive psychological factors, such as proactive coping and optimism.

factors, including psychological factors (2). Examples of psychological factors are proactive coping, self-efficacy, extraversion, optimism, passive coping, neuroticism and pessimism.

In stroke, specifically, previous research has revealed associations between psychological factors and participation or quality of life (3–5). Research has shown that proactive coping, self-efficacy, extraversion and optimism are related to better outcomes, whereas passive coping, neuroticism and pessimism are related to worse outcomes (3–7). These strong associations with outcome highlight the importance of gaining knowledge about the properties of these psychological factors, such as their stability after stroke. Knowledge about temporal stability can be used to optimize the timing of psychological assessments and to gain knowledge about whether these factors can be influenced in such a way that outcome can be optimized.

In healthy people psychological factors, especially personality traits, are generally regarded as stable (8). Most studies confirm the temporal stability among psychological factors in healthy adults (9–13), but a few studies have reported changes over time, in e.g. optimism (14) and neuroticism (10, 12). Studies in patients with stroke and similar neurological conditions, such as traumatic brain injury (TBI), have shown mixed findings. Regarding personality traits, inconsistent results across and within factors have been found, and specific evidence remains scarce for patients with stroke (11, 12, 15–17). These inconsistencies are also found with regard to coping (3, 17, 18), possibly also because terminology for subtypes of coping styles differ widely (19).

M. L. M. Wijenberg et al.

These inconsistencies and limited findings in patients with stroke show the need for further exploration of the temporal stability of psychological factors. Therefore, the main aim of the current study was to explore whether psychological factors, namely proactive coping, self-efficacy, extraversion, optimism, passive coping, neuroticism and pessimism, are stable over the first 2 years post-stroke. Furthermore, we explored possible clustering of these psychological factors.

METHODS

Design and procedure

The current study was part of the Restore4Stroke cohort study, a multicentre prospective longitudinal cohort study in which patients with stroke were recruited on admission to the acute stroke unit at 1 of 6 general hospitals in the Netherlands and were followed up for 2 years after their stroke. Details of the study procedures are given elsewhere (20). The medical ethics committees of all participating hospitals provided approval. Participants were recruited between March 2011 and March 2013 and all provided written informed consent.

After obtaining participants' informed consent, information on stroke-related factors (e.g. type of stroke, lateralization, stroke severity, activities of daily living (ADL) dependency) was extracted from the hospital database. Information on demographic factors was obtained from the patient or family members (within the first week post-stroke). At 2 months post-stroke, participants completed self-report questionnaires regarding psychological factors and underwent cognitive screening with a trained research assistant. At 2 years post-stroke, participants completed the same self-report questionnaires regarding psychological factors. The questionnaires were administered on computer or paper and were guided by a research assistant if necessary.

Participants

Inclusion criteria of the Restore4Stroke cohort study were: ≥ 18 years old, having a diagnosis of stroke (ischaemic or intracerebral haemorrhagic lesion) clinically confirmed by the treating neurologist and stroke onset within the last 7 days.

Exclusion criteria were: (*i*) having a serious other condition that could be expected to influence the study outcomes (e.g. cancer or dementia); (*ii*) having already been dependent regarding activities of daily living (ADL) before the stroke, as defined by a Barthel Index (BI) score ≤ 17 ; (*iii*) having insufficient command of the Dutch language to understand and complete the questionnaires, based on clinical judgment; and (*iv*) experiencing cognitive decline before the stroke, as defined by a score ≥ 1 on the Heteroanamnesis List Cognition (HLC). The HLC is a 5-item questionnaire, completed by the patient's spouse, assessing premorbid cognitive dysfunctioning on 5 cognitive domains. Participants who completed the questionnaires for psychological factors at 2 years post-stroke were selected for the current study.

Measures

Details and references for all measures are described elsewhere (20).

Demographic and stroke-related factors. Data regarding age, sex, education and marital status (single or in a relationship)

were collected. The highest completed level of education was recorded according to the 7-point Verhage system, classifying the Dutch education system into 7 categories ranging from no primary school completed ("1") to completion of a higher educational level (such as college ("6") and university ("7")). Stroke-related factors, such as type (ischaemic, haemorrhagic or unknown/other), lateralization (left, right, cerebellar, brainstem or unknown/other), severity (assessed by the National Institutes of Health Stroke Scale (NIHSS)) and discharge destination (home, or (geriatric) rehabilitation) were collected. ADL was measured with the Barthel Index at day 4 post-stroke and cognitive functioning was measured with the Montreal Cognitive Assessment (MoCA) at 2 months post-stroke.

Psychological factors. All questionnaires to assess psychological factors are reliable and valid in patients with stroke (20). Proactive coping competencies were assessed with the Utrecht Proactive Coping Competence scale (UPCC). The 1-month test–retest reliability is 0.72 (21). The questionnaire consists of 21 items scored on 4-point scale ranging from "not very competent" to "competent". A mean score is calculated, ranging from 1 to 4, with a higher score indicating a higher level of proactive coping competencies.

Self-efficacy was assessed with the General Self-Efficacy Scale (GSES). The 7-week test-retest reliability is 0.72 (22). The scale consists of 10 items scored on a 4-point scale ranging from "not at all true" to "exactly true". A sum score is calculated, ranging from 10 to 40, with a higher score indicating a higher level of self-efficacy.

Extraversion and neuroticism were assessed with 2 subscales of the Eysenck Personality Questionnaire Revised Short Scale (EPQ-SS). Both scales consist of 12 items with a dichotomous (yes/no) response option. The 6-month test-retest reliability is 0.85 for optimism and 0.70 for neuroticism (23). A sum score is calculated, ranging from 0 to 12, with a higher score indicating a higher level of extraversion or neuroticism, respectively.

Optimism and pessimism were assessed with the Life Orientation Test Revised (LOT-R). The 4-month test-retest reliability is 0.68 (24). It consists of 10 items, with 3 items measuring optimism, 3 items measuring pessimism and 4 filler items. The items are scored on a 5-point scale ranging from "strongly disagree" to "strongly agree". A score is calculated per subscale, ranging from 0 to 12, with a higher score indicating a higher level of optimism or pessimism, respectively.

Passive coping was assessed with the passive reaction pattern subscale of the Utrecht Coping List (UCL). The 6-week testretest reliability is 0.84 for the passive reaction pattern subscale (25). The questionnaire consists of 7 items, scored on a 4-point scale ranging from "seldom" to "very often". A sum score is calculated, ranging from 7 to 28, with a higher score indicating a higher level of passive coping.

Statistical analyses

Preparatory analyses. All data analyses were performed using SPSS Statistics 24.0 for Windows (IBM Corp., Armonk, NY, USA). Missing data were inspected and imputed with the mean value within the corresponding subscale if at least 80% of the participant's data of the specific questionnaire was available. For the GSES, the criterion suggested by the manual of at least 70% non-missing data was used. Data were inspected for normality and outliers. In case of non-normality (skewness or kurtosis value <-1 or >+1), data were transformed. Skewness to the right was resolved with a logarithmic transformation. A quadratic transformation was executed to resolve skewness

2

JRM

to the left. All analyses were performed using an alpha level of 0.05.

Baseline differences. Group differences on baseline characteristics between the included and excluded participants were analysed using Pearson χ^2 tests for nominal variables including all available levels, independent *t*-tests (*T*) for normally distributed continuous variables, and non-parametric tests (Mann-Whitney *U* test) for continuous variables in case of non-normality.

Changes over time. To reveal changes over time, paired samples *t*-tests were performed. For further inspection, changes over time were categorized into 3 categories: "worse", "equal" and "better". Hereby, we used a change of 0.5SD as pragmatic indicator of change of the psychological variable score over time, as suggested by Norman et al. (26). For the variables proactive coping, self-efficacy, extraversion and optimism, an increase of more than 0.5 SD over time was considered an adaptive change ("better") (see Introduction). For the variables passive coping, neuroticism and pessimism an increase of more than 0.5 SD over time was considered a maladaptive change ("worse") (see Introduction). A change of less than 0.5 SD over time was considered as no change over time ("equal"). Effect sizes were calculated using Glass's delta (27).

Correlation and factor analyses. To explore associations among the psychological variables, Pearson correlations were calculated and interpreted as suggested by Evans (28) (≤ 0.19 very weak; 0.20-0.39 weak; 0.40-0.59 moderate; 0.60-0.79 strong; ≥ 0.80 very strong). To explore dimensions among the psychological variables, exploratory factor analysis (EFA) (including the extraction methods principal axis factoring and maximum likelihood estimation) was performed using data from participants who completed all questionnaires at both time-points (n=282). Factor extraction was based on multiple criteria, as suggested by Osborne & Costello (29). The K1 criterion (eigenvalue ≥ 1.0), the Scree plot, small residual correlations ([-0.1; 0.1]) and a non-significant goodness-of-fit test (based on maximum likelihood estimation) were used as 4 indicators for extraction. If the criteria suggested multiple solutions for factor extraction, e.g. a 1-factor structure on the basis of the Scree plot and a 2-factor structure on the basis of the K1 crite-

Table I. Patients' characteristics at baseline

rion, all suggestions of factor extraction were executed and the best solution, indicated by smaller residual correlations and a non-significant goodness-of-fit test, was chosen. Only factor loadings greater than 0.30 were displayed. Because we expected the factors to be correlated, Oblimin rotation was preferred over Varimax rotation.

RESULTS

Sample

Of the 395 participants in the Restore4Stroke cohort, 324 (82.0%) provided data on psychological factors at 2 years post-stroke and were included in the current study. Furthermore, depending on the psychological factor, imputation was performed for 0.0-9.6% of participants. Baseline characteristics of both the included group (n=324) and excluded group (n=71)and significant differences between them are shown in Table I. The participants in the included group were significantly younger, less often highly educated and more often involved in a relationship in comparison with the excluded group. Regarding stroke-related factors, the included group had a significantly higher level of cognitive functioning 2 months post-stroke and a significantly higher level of independence in ADL 4 days post-stroke in comparison with the excluded group.

Change of psychological factors over time

Table II provides the results of the temporal stability analysis of psychological factors. All psychological factors except passive coping showed significant changes over time. At 2 years post-stroke, participants reported less proactive coping, lower self-efficacy, less extraversion, less optimism, more neuroticism

	Included group $(n = 324)$	Excluded group (n=71)	Test statistic $\chi^2/T/U$	<i>p</i> -value
Demographic factors				
Male, %	64.8	64.8	0.00	1.00
Age, years, mean (SD)	65.9 (12.1)	70.4 (14.0)	2.79	< 0.01
Marital status: in a relationship, %	71.3	56.3	6.05	0.01
Higher educational level, %	25.4 ^a	32.8 ^b	19.68	< 0.01
Stroke-related factors				
Ischaemic, %	92.3	97.2	2.23	0.33
Left hemisphere, %	39.0 ^a	45.1	8.05	0.09
Severity of stroke (NIHSS), mean (SD)	2.7 (3.2)	3.3 (3.3)	9,975.00	0.08
No stroke symptoms (NIHSS 0), %	24.4	21.1		
Minor stroke symptoms (NIHSS 1-4), %	57.1	50.7		
Moderate stroke symptoms (NIHSS 5–12), %	16.4	25.4		
Moderate to severe stroke symptoms (NIHSS 13), %	2.1	2.8		
ADL 4 days post-stroke (BI), mean (SD)	17.0 (4.8)	16.1 (4.9)	9,575.50	0.02
Cognitive functioning 2 months post-stroke (MoCA)	23.8 (3.7) ^c	21.9 (5.4) ^d	5,072.50	< 0.05
Cognitively impaired (MoCA \leq 25), %	66.3	75.6		
Destination after discharge from hospital			4.52	0.10
Home, %	72.2	62.0		
(Geriatric) rehabilitation, %	27.8	38.0		

^an = 323, ^bn = 61, ^cn = 306, ^dn = 41. T: independent *t*-test; U: Mann-Withney *U* test; NIHSS: National Institutes of Health Stroke Scale; ADL: activities of daily living; BI: Barthel Index; MoCA: Montreal Cognitive Assessment; SD: standard deviation.

M. L. M. Wijenberg et al.

4

Table II. Temporal stability of psychological factors

			2 months post-stroke 2 years post-stroke				Changes over time 0.5 SD ^a			
Psychological factors	Score range	п	Mean (SD)	Mean (SD)	t	Effect size	%worse	%equal	%better	Correlations ^b
Proactive coping	[1-4]	301	3.1 (0.5)	2.9 (0.7)	6.83**	-0.42	46.8	33.6	19.6	0.55**
Self-efficacy	[10-40]	301	31.5 (6.4)	30.8 (6.4)	2.20* ^c	-0.11	28.6	51.5	19.9	0.52** ^c
Extraversion	[0-12]	305	7.2 (3.2)	6.3 (3.3)	5.64**	-0.28	34.1	51.1	14.8	0.64**
Optimism	[0-12]	298	8.2 (2.1)	6.9 (2.1)	9.33**	-0.58	41.6	47.7	10.7	0.42**
Passive coping	[7-28]	304	10.5 (2.8)	10.9 (3.2)	-0.87 ^c	0.12	28.9	48.4	22.7	0.61** ^c
Neuroticism	[0-12]	305	3.6 (3.1)	4.4 (3.5)	-5.12**	0.27	34.8	47.5	17.7	0.64**
Pessimism	[0-12]	294	4.4 (2.8)	4.9 (2.3)	-3.38**	0.19	35.0	43.2	21.8	0.49**

^a% worse, % equal and % better refer to categories of 0.5 SD change of the psychological variable score over time.

^bCorrelations between 2 months post-stroke and 2 years post-stroke.

^cBased on transformed data.*p < 0.05, **p < 0.01.

and more pessimism in comparison with 2 months post-stroke. The largest effect sizes were found for optimism (Glass's $\Delta = -0.58$) and proactive coping (Glass's $\Delta = -0.42$). Furthermore, looking at changes over time ("worse", "equal", "better"), 28.6–46.8% participants showed changes in a negative direction and 10.7–22.7% improved across all psychological factors.

Associations among psychological factors

For all psychological factors moderate or strong correlations were found between scores at 2 months and 2 years post-stroke, ranging between 0.42 and 0.64 (Table II). In Table III correlations among the various psychological factors within each time-point are reported.

At 2 months post-stroke, all psychological factors correlated significantly with each other, except for extraversion with neuroticism. The correlations ranged between very weak to strong. The highest correlation was found between proactive coping and self-efficacy (r=0.62). At 2 years post-stroke, all psychological factors correlated significantly with each other. The correlations ranged between weak and strong. The highest correlation was again found between proactive coping and self-efficacy (r=0.71). Overall, correlations at 2 years were somewhat stronger compared with 2 months.

Dimensions among psychological factors

The correlations among the psychological variables at both time-points were best described by the extraction of 2 factors using principal axis factoring as EFA extraction method (Table IV). After factor extraction and oblimin rotation, the factor correlation was 0.58 and -0.70 at 2 months post-stroke and 2 years post-stroke, respectively. The pattern matrices of both time-points, consisting of the factor loadings after oblimin rotation, are shown in Table V. For interpretation purposes, factor loadings on factor 1 at 2 months post-stroke were inverted.

At 2 months post-stroke, factors 1 and 2 are labelled as "maladaptive psychological factor" and "adaptive

 Table III. Correlations of psychological measures at 2 months post-stroke (below the diagonal) and at 2 years post-stroke (above the diagonal)

Psychological factors	1	2 ^a	3 ^a	4	5	6	7
1. Proactive coping	-	0.71**	0.50**	0.47**	-0.54**	-0.46**	-0.30**
2. Self-efficacy ^a	0.62**	-	0.45**	0.46**	-0.56**	-0.47**	-0.34**
3. Extraversion	0.27**	0.27**	-	0.43**	-0.40**	-0.28**	-0.29**
4. Optimism	0.41**	0.44**	0.21**	-	-0.56**	-0.53**	-0.37**
5. Passive coping ^a	-0.46**	-0.39**	-0.19**	-0.41**	-	0.68**	0.34**
6. Neuroticism	-0.40**	-0.31**	-0.09	-0.39**	0.59**	-	0.38**
7. Pessimism	-0.15**	-0.24**	-0.14*	-0.28**	0.26**	0.26**	-

Correlation coefficient is Pearson's correlation. ^aAt 2 months post-stroke based on transformed data due to non-normality.*p<0.05, **p<0.01.

Table IV. Exploration of factor structures at 2 months post-stroke and 2 years post-stroke

Criterion for factor extraction	Two months pos	Two months post-stroke T		Two years post-stroke	
N of extracted factors	1	2	1	2	
Eigenvalues	3.03	1.02	3.76	0.86	
K1 criterion (eigenvalues ≥ 1.0)					
Scree plot	\checkmark		\checkmark		
Residual correlations	[-0.06; 0.14]	[-0.07; 0.09]	[-0.07; 0.14]	[-0.04; 0.06]	
Small [-0.1; 0.1]					
<i>p</i> -value, goodness-of-fit test	0.00	0.08	0.00	0.10	
Non-significant					

Bold indicates that the criterion is met.

5

Table V. Factor loadings based on exploratory factor analysis after Oblimin rotation

	Two months post-stroke		Two years post-stroke ^a	
	Factor 1 "Maladaptive PF" ^b	Factor 2 "Adaptive PF"	Factor 1 "Adaptive PF"	Factor 2 "Maladaptive PF"
Proactive coping		0.61	0.85	
Self-efficacy ^c		0.81	0.77	
Extraversion		0.37	0.58	
Optimism	-0.36	0.33		-0.51
Passive coping ^c	0.72			0.68
Neuroticism	0.86			0.97
Pessimism				0.35
Eigenvalues	3.03	1.02	3.76	0.86
Explained variance, %	43	43.55 54.84		
Factor correlation	-0	-0.58 -0.70		

Factor loadings < 0.3 are suppressed.

^aForced 2-factor analysis.

^bFactor correlation and loadings are inverted for interpretation purposes. ^cBased at 2 months post-stroke on transformed data due to non-normality.

PF: psychological factor.

psychological factor", respectively. Passive coping and neuroticism had positive loadings on the maladaptive factor, whereas optimism had a negative loading on this factor. Proactive coping, self-efficacy, extraversion and optimism had positive loadings on the adaptive factor. Pessimism did not load on any factor. The explained variance of the 2 factors together was 43.6%.

At 2 years post-stroke, factors 1 and 2 are labelled as "adaptive psychological factor" and "maladaptive psychological factor", respectively. Proactive coping, self-efficacy and extraversion had positive loadings on the adaptive factor. Passive coping, neuroticism and pessimism had positive loadings on the maladaptive factor. Furthermore, optimism had a negative loading on this factor. The explained variance of the 2 factors together was 54.8%.

DISCUSSION

This study found that scores on measures of psychological factors changed during the first 2 years poststroke. The psychological factors were moderately to strongly correlated with each other and over time and clustered at both time-points into 2 factors: an "adaptive psychological factor" (proactive coping, self-efficacy, extraversion) and a "maladaptive psychological factor" (passive coping and neuroticism). Scores on all adaptive psychological factors decreased over time, whereas scores on maladaptive psychological factors increased or remained stable (passive coping) over time. The added value of our study is that we evaluated the temporal stability of multiple psychological factors simultaneously in a large cohort of patients with stroke using a longitudinal design.

To the best of our knowledge, the finding of a negative temporal impact across all psychological factors post-stroke has not been demonstrated previously. As mentioned in the introduction, previous research on temporal stability of psychological factors in stroke and other populations showed inconsistent results across and within psychological factors, possibly due to the use of different measures, time-points and designs.

Strikingly, even though the sample consisted mainly of patients with a mild stroke overall changes in a negative direction were found. In comparing our results with previous findings regarding temporal stability of psychological factors in healthy populations, we found different patterns of changes over time. Most studies assessing psychological factors in healthy adults showed temporal stability (9–14), whereas our data showed temporal changes in a negative direction across all psychological factors. This suggests that the occurrence of a stroke could be a possible cause of the observed negative changes over time.

The occurrence of stroke results in negative changes regarding emotional, cognitive and energy resources, which could provide an explanation for the observed negative change of psychological factors over time. After a stroke, many patients suffer from depression (30), cognitive complaints (31) and disabling fatigue (32), increasing their burden and calling on their reserves of resilience. The observed negative changes in psychological factors could be concomitant with these negative emotional, cognitive and energy changes. In fact, it was shown that neuroticism is concomitant with depressive symptoms (33), and that cognitive complaints in patients with traumatic brain injury (TBI) are associated with the use of maladaptive coping styles (7). In our sample, 66% were considered to be cognitively impaired (MoCA score \leq 25) at 2 months post-stroke. Furthermore, Wu et al. (32) proposed a biopsychosocial model including psychological factors as an explanation for fatigue after stroke. Taken together, stroke is associated with negative changes in emotional, cognitive or energy resources. These consequences of stroke may form

an explanation for the negative changes observed in psychological factors.

Although the occurrence of stroke provides a possible explanation for the significant differences over time on all measures except for passive coping, it remains unclear whether damage to the brain (i.e. the stroke) or the fact that a stroke is a negative life event (such as a cardiac event) or the chronic character (as in other diseases, such as diabetes) or their combination may induce changes in psychological factors. From earlier studies we know that patients with acquired brain injury make more use of passive reactions and less use of problem-solving coping styles than the general population (7), which may be related to the damage to the brain. From studies on survivors of a cardiac arrest we know that the impact of the event plays an important role in the quality of life in the long term (34); therefore distress may also influence psychological functioning. A study on chronic diseases (35) also found changes over time in extraversion and neuroticism after the onset of chronic diseases, such as heart disease, respiratory disease and stroke. Future studies are needed to relate the observed negative changes to stroke, brain damage, a negative life event, chronic character of the disease or their combination.

Furthermore, one could argue whether these changes comply with the minimal clinically relevant change of these measures and therefore represent clinically relevant changes and/or changes due to measurement error. Because this information is not available, we chose 0.5 SD as pragmatic indicator of change, as suggested by Norman et al. (26). Future research should provide the minimal clinically relevant change per measure to investigate whether the changes observed in this study represent clinically relevant changes. Even if the observed changes do not reflect clinically relevant changes and thus suggest temporal stability, our findings imply, at the very least, that the psychological factors investigated do not improve naturally to a more beneficial level and, consequently, do not foster improved outcomes in terms of participation and quality of life.

To examine whether psychological factors can be modified by treatment programmes, systematic reviews and meta-analyses provide some evidence for the ability to change depression, anxiety, self-efficacy and coping by means of psychological therapy, to possibly improve the outcomes of patients with stroke in terms of quality of life and participation (36–40). Given these positive findings related to the modifiability of some psychological factors, treatment could be aimed at enhancing adaptive psychological factors and limiting maladaptive psychological factors in order to improve participation and quality of life of patients with stroke. However, to provide more insight for the development of such focused and personalized treatment, future research should reveal which patients are at risk of the negative changes over time regarding psychological factors.

Another implication is related to the occurrence of 2 clusters, namely an "adaptive psychological factor" (proactive coping, self-efficacy, extraversion) and a "maladaptive psychological factor" (passive coping and neuroticism). This could suggest the use of a single measure of adaptive psychological factors and a single measure of maladaptive psychological factors for subsequent studies. However, more research is needed to replicate these findings and provide direction for the selection or development of such measures. Also it should be noted that at 2 months post-stroke, pessimism did not load on the extracted factors, whereas optimism loaded on both factors. At 2 years post-stroke, pessimism did load on the adaptive psychological factor, but optimism, although inverted, also loaded on this maladaptive factor and not, as expected, on the adaptive factor.

While interpreting the results of our study, the following limitations should be taken into account. First, the homogeneous sample could limit the generalizability of our results to the entire stroke population or to other patient populations. Most patients in our study had a minor ischaemic stroke. An explanation for this homogeneity is that mild stroke comprises the largest group of stroke patients and patients with a severe ischaemic stroke or a haemorrhagic stroke are less often present, are less often referred to general hospitals, have greater difficulty understanding the questionnaires or study instructions and are less able to provide informed consent within the first week (4). Future research could investigate the temporal stability of psychological factors in patients with a more severe or haemorrhagic stroke. Furthermore, it would be interesting to examine whether, in other patient populations with mild brain injuries, such as mild TBI, or in other chronic diseases with a sudden onset, the same negative effect of time is seen to reveal whether these changes are specific for a mild brain injury (stroke or mild TBI) or related to a more general cause, such as the onset of a disabling chronic disease or occurrence of other significant (health-related) life events. Secondly, psychological factors were measured at 2 time-points, 22 months apart. To provide evidence for the existence of linear or non-linear time effects and to reveal timerelated changes within this time period psychological factors should ideally be assessed at more time-points. Thirdly, time-dependent relationships with regard to depression, anxiety and other psychological factors, such as locus of control, sense of coherence and resi-

6

Journal of Rehabilitation Medicine

lience were not considered in this study. This limits any conclusion regarding these other psychological factors or the existence of a mediating role, which might be played by a more general negative emotional state (33).

Despite these limitations, we found, in a large prospective longitudinal multicentre cohort study across 7 different psychological factors in patients with stroke, changes toward less favourable scores, which may be associated with worse outcome, such as participation and quality of life. Clinicians should attend to adaptive psychological factors (e.g. proactive coping, selfefficacy, extraversion) and maladaptive psychological factors (e.g. passive coping and neuroticism) during long-term care.

ACKNOWLEDGEMENT

The presented study is part of the Restore4Stroke Research Program, which is supported financially by VSBfund (grant no. 89000004) and coordinated by ZonMw (the Dutch Organization for Health Research and Development).

The authors have no conflicts of interest to declare.

REFERENCES

- 1. Benjamin EJ, Blaha MJ, Chiuve SE, Cushman M, Das SR, Deo R, et al. Heart disease and stroke statistics – 2017 update: a report from the American Heart Association. Circulation 2017; 135: e146–e603.
- Geyh S, Peter C, Müller R, Bickenbach JE, Kostanjsek N, Üstün BT, et al. The personal factors of the international classification of functioning, disability and health in the literature – a systematic review and content analysis. Disabil Rehabil 2011; 33: 1089–1102.
- Darlington A-SE, Dippel DWJ, Ribbers GM, van Balen R, Passchier J, Busschbach JJV. Coping strategies as determinants of quality of life in stroke patients: a longitudinal study. Cerebrovasc Dis 2007; 23: 401–407.
- van Mierlo M, van Heugten C, Post MWM, Hoekstra T, Visser-Meily A. Trajectories of health-related quality of life after stroke: results from a one-year prospective cohort study. Disabil Rehabil 2017: 1–10.
- van Mierlo ML, Schröder C, van Heugten CM, Post MWM, de Kort PLM, Visser-Meily JMA. The influence of psychological factors on health-related quality of life after stroke: a systematic review. Int J Stroke 2014; 9: 341–348.
- Kim S, Kim J, Stewart R, Kang H, Kim S, Shin I, et al. Influences of personality traits on quality of life after stroke. Eur Neurol 2013; 69: 185–192.
- Wolters Gregório G, Ponds RWHM, Smeets SMJ, Jonker F, Pouwels CGJG, Verhey FR, et al. Associations between executive functioning, coping, and psychosocial functioning after acquired brain injury. Br J Clin Psychol 2015; 54: 291–306.
- McCrae RR, Costa PT. Personality in adulthood: a five-factor theory perspective: Guilford Press; 2003.
- 9. Carver CS, Scheier MF, Segerstrom SC. Optimism. Clin Psychol Rev 2010; 30: 879–889.
- Karsten J, Penninx BWJH, Riese H, Ormel J, Nolen WA, Hartman CA. The state effect of depressive and anxiety disorders on big five personality traits. J Psychiatr Res 2012; 46: 644–650.
- 11. Kurtz JE, Putnam SH, Stone C. Stability of normal personality traits after traumatic brain injury. J Head Trauma

Temporal stability of psychological factors post-stroke

Rehabil 1998; 13: 1-14.

- Leonhardt A, Schmukle S, Exner C. Evidence of big-five personality changes following acquired brain injury from a prospective longitudinal investigation. J Psychosom Res 2016; 82: 17–23.
- Nielsen MB, Knardahl S. Coping strategies: a prospective study of patterns, stability, and relationships with psychological distress. Scand J Psychol 2014; 55: 142–150.
- 14. Chopik WJ, Kim ES, Smith J. Changes in optimism are associated with changes in health over time among older adults. Soc Psychol Personal Sci 2015; 6: 814–822.
- Lannoo E, De Deyne C, Colardyn F, De Soete G, Jannes C. Personality change following head injury: assessment with the neo five-factor inventory. J Psychosom Res 1997; 43: 505–511.
- 16. Sabariego C, Barrera AE, Neubert S, Stier-Jarmer M, Bostan C, Cieza A. Evaluation of an ICF-based patient education programme for stroke patients: a randomized, single-blinded, controlled, multicentre trial of the effects on self-efficacy, life satisfaction and functioning. Br J Health Psychol 2013; 18: 707–728.
- Scheenen ME, van der Horn HJ, de Koning ME, van der Naalt J, Spikman JM. Stability of coping and the role of self-efficacy in the first year following mild traumatic brain injury. Soc Sci Med 2017; 181: 184–190.
- King RB, Shade-Zeldow Y, Carlson CE, Feldman JL, Philip M. Adaptation to stroke: a longitudinal study of depressive symptoms, physical health, and coping process. Top Stroke Rehabil 2002; 9: 46–66.
- Wolters Gregório G, Brands I, Stapert S, Verhey FR, van Heugten CM. Assessments of coping after acquired brain injury. J Head Trauma Rehabil 2014; 29: E30–E42.
- van Mierlo ML, van Heugten CM, Post MWM, Lindeman E, de Kort PLM, Visser-Meily JMA. A longitudinal cohort study on quality of life in stroke patients and their partners: Restore4Stroke cohort. Int J Stroke 2014; 9: 148–154.
- Bode C, Thoolen B, de Ridder D. Measuring proactive coping. Psychometric characteristics of the Utrecht Proactive Coping Competence scale (UPCC). Psychologie & Gezondheid 2008; 36: 81–91.
- Leganger A, Kraft P, Røysamb E. Perceived self-efficacy in health behaviour research: Conceptualisation, measurement and correlates. Psychol Health 2000; 15: 51–69.
- Sanderman R, Arrindell W, Ranchor A. [Measuring personality aspects with the Eysenck Personality Questionnaire (EPQ)]. 2nd edn. UMCG/Rijksuniversiteit Groningen, Research Institute SHARE; 2012 (in Dutch).
- Scheier MF, Carver CS, Bridges MW. Distinguishing optimism from neuroticism (and trait anxiety, self-mastery, and self-esteem): a reevaluation of the Life Orientation Test. J Pers Soc Psychol 1994; 67: 1063–1078.
- 25. Turner H, Bryant-Waugh R, Peveler R, Bucks RS. A Psychometric Evaluation of an English Version of the Utrecht Coping List. Eur Eat Disord Rev 2012; 20: 339–342.
- Norman GR, Sloan JA, Wyrwich KW. Interpretation of changes in health-related quality of life. Med Care 2003; 41: 582–592.
- 27. Glass GV. Primary, Secondary, and meta-analysis of research. Educ Res 1976; 5: 3.
- Evans JD. Straightforward statistics for the behavioral sciences. Pacific Groove: Brooks/Cole Publishing Co.; 1996.
- 29. Osborne JW, Costello AB. Best practices in exploratory factor analysis: four recommendations for getting the most from your analysis. Pan-Pacific Manag Rev 2009; 12: 131–146.
- Ayerbe L, Ayis S, Wolfe CDA, Rudd AG. Natural history, predictors and outcomes of depression after stroke: systematic review and meta-analysis. Br J Psychiatry 2013; 202: 14–21.
- Eskes GA, Lanctôt KL, Herrmann N, Lindsay P, Bayley M, Bouvier L, et al. Canadian Stroke best practice recommendations: mood, cognition and fatigue following stroke

M. L. M. Wijenberg et al.

Journal of Rehabilitation Medicine

JRN

practice guidelines, Update 2015. Int J Stroke 2015; 10: 1130-1140.

- 32. Wu S, Mead G, Macleod M, Chalder T. Model of understanding fatigue after stroke. Stroke 2015; 46: 893–898.
- Sherry SB, Mackinnon SP, Macneil MA, Fitzpatrick S. Discrepancies confer vulnerability to depressive symptoms: a three-wave longitudinal study. J Couns Psychol 2013; 60: 112–126.
- Moulaert V, Wachelder EM, Verbunt JA, Wade DT, van Heugten CM. Determinants of quality of life in survivors of cardiac arrest. J Rehabil Med 2010; 42: 553–558.
- Jokela M, Hakulinen C, Singh-Manoux A, Kivimäki M. Personality change associated with chronic diseases: pooled analysis of four prospective cohort studies. Psychol Med 2014; 44: 2629–2640.
- Hildebrand MW. Effectiveness of interventions for adults with psychological or emotional impairment after stroke: an evidence-based review. Am J Occup Ther 2015; 69:

6901180050p1-9.

- Jones F, Riazi A. Self-efficacy and self-management after stroke: a systematic review. Disabil Rehabil 2010; 33: 797–810.
- Mehta S, Pereira S, Janzen S, McIntyre A, McClure A, Teasell RW. Effectiveness of psychological interventions in chronic stage of stroke: a systematic review. Top Stroke Rehabil 2012; 19: 536–544.
- 39. Stalder-Lüthy F, Messerli-Bürgy N, Hofer H, Frischknecht E, Znoj H, Barth J. Effect of psychological interventions on depressive symptoms in long-term rehabilitation after an acquired brain injury: a systematic review and meta-analysis. Arch Phys Med Rehabil 2013; 94: 1386–1397.
- 40. Visser MM, Heijenbrok-Kal MH, van't Spijker A, Lannoo E, Busschbach JJV, Ribbers GM. Problem-solving therapy during outpatient stroke rehabilitation improves coping and health-related quality of life. Stroke 2016; 47: 135–142.

www.medicaljournals.se/jrm

8